Author’s response to reviews

Title: 4q27 deletion and 7q36.1 microduplication in a patient with multiple malformations and hearing loss: a case report

Authors:

Maolan Wu (wumaolan@csu.edu.cn)
Xiangrong Zheng (xrzheng@csu.edu.cn)
Xia Wang (wangxiachen@163.com)
Guoyuan Zhang (13974870175@163.com)
Jian Kuang (656850280@qq.com)

Version: 2 Date: 09 Feb 2020

Author’s response to reviews:

1. Availability of data and materials.

Answer: Due to the current outbreak of novel coronavirus (2019-nCoV), All schools in china are required to delay the opening of the 2020 spring semester until the end of February, and the holiday of most companies are extended. The resumption date will be confirmed later in accordance with the development of the fight against the novel coronavirus epidemic. We can not get in touch with the Lab to get the data of CMA( Chromosomal microarray analysis).

2. Authors’ contributions

Please note the order of initials when referring to an author in the Authors' contributions should be [first initial of first name][first initial of last name]. Please amend the order of initials in regard to each author as appropriate.

Answer: The order of initials in regard to each author have been amended in the Authors’ contributions.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

Answer: We upload the manuscript as a single, final, clean version. The table and figures are also clean versions.